Author's response to reviews

Title: Study protocol: The InterHerz project- A web-based psychological treatment for cardiac patients with depression: a randomized controlled trial

Authors:

Nadine Messerli-Burgy (nadine.messerli@psy.unibe.ch)
Jürgen Barth (mail@juergen-barth.de)
Thomas Berger (thomas.berger@ptp.unibe.ch)

Version: 2 Date: 28 November 2012

Author's response to reviews: see over